Author's response to reviews

Title: Prescription of antibiotics and anxiolytics/hypnotics to asthmatic patients in general practice: A cross-sectional study based on French and Italian prescribing data

Authors:

David Darmon (david.darmon@unice.fr)
Laurent Laforest (laurent.laforest@univ-lyon1.fr)
Eric Van ganse (eric.van-ganse@univ-lyon1.fr)
Ferdinando Petrazzuoli (ferdinando.petrazzuoli@gmail.com)
Chris van Weel (C.vanWeel@elg.umcn.nl)
Laurent Letrilliart (laurent.letrilliart@wanadoo.fr)

Version: 4
Date: 25 December 2014

Author's response to reviews:

Ref.: MS: 8214463391434185

Manuscript Title: Prescription of antibiotics and anxiolytics/hypnotics to asthmatic patients in general practice: A cross-sectional study based on French and Italian prescribing data.

David Darmon Laurent Laforest Eric Van ganse Ferdinando Petrazzuoli Chris van Weel and Laurent Letrilliart

Dear Editor,

Please find enclosed our revised manuscript, entitled “Prescription of antibiotics and anxiolytics/hypnotics to asthmatic patients in general practice: A cross-sectional study based on French and Italian prescribing data.”

We revised our manuscript according to your last comments. Thus, a native-English speaker with scientific expertise of the professional language editing service Edanz (ref: G1412-8302) copyedited our manuscript to improve its English style.

Sincerely yours,
Best regards,

David DARMON, on behalf of the authors

Corresponding author: David DARMON Département d'enseignement et de recherche en médecine générale, UFR médecine, Université de Nice Sophia-Antipolis, 28, avenue Valombreuse 06 107 Nice Cedex 02, France